Author's response to reviews

Title: Prevalence and determinants of tobacco use amongst junior collegiates in twin cities of western Nepal: A cross-sectional, questionnaire-based survey

Authors:

Chandrashekhar T Sreeramareddy (chandrashekharats@yahoo.com)
P V Kishore (pandu_vki@yahoo.com)
Jagadish Paudel (paudeljg@hotmail.com)
Ritesh G Menezes (mangalore971@yahoo.co.in)

Version: 4 Date: 14 March 2008

Author's response to reviews:

The references were corrected.

Figures were uploaded as separate files.

Thank you